Check for updates

Citation: Lorenz C, Azevedo TS, Virginio F, Aguiar BS, Chiaravalloti-Neto F, Suesdek L (2017) Impact of environmental factors on neglected emerging arboviral diseases. PLoS Negl Trop Dis 11(9): e0005959. https://doi.org/10.1371/journal. pntd.0005959

Editor: Marilia Sá Carvalho, Fundacao Oswaldo Cruz, BRAZIL

Received: May 30, 2017

Accepted: September 12, 2017

Published: September 27, 2017

Copyright: © 2017 Lorenz et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: CL was supported by funding from Fundação de Amparo a Pesquisa do Estado de São Paulo (FAPESP) (2013/05521-9). FV was supported by Coordenação de Aperfeiçoamento de Pessoal de Nível Superior (CAPES) (23038.005.274/2011-24 and 1275/2011). LS was supported by Conselho Nacional de Desenvolvimento Científico e Tecnológico (CNPq) RESEARCH ARTICLE

Impact of environmental factors on neglected emerging arboviral diseases

Camila Lorenz¹*, Thiago S. Azevedo², Flávia Virginio¹, Breno S. Aguiar³, Francisco Chiaravalloti-Neto^{3©}, Lincoln Suesdek^{1,4©}

1 Department of Parasitology, Instituto Butantan, Sao Paulo, Sao Paulo, Brazil, 2 Department of Geography, Universidade Estadual Paulista, Rio Claro, Sao Paulo, Brazil, 3 Department of Epidemiology, School of Public Health, Universidade de São Paulo, Sao Paulo, Brazil, 4 Instituto de Medicina Tropical, Sao Paulo, São Paulo, Brazil

These authors contributed equally to this work.

* cammillalorenz@gmail.com

Abstract

Background

Brazil is a tropical country that is largely covered by rainforests and other natural ecosystems, which provide ideal conditions for the existence of many arboviruses. However, few analyses have examined the associations between environmental factors and arboviral diseases. Thus, based on the hypothesis of correlation between environment and epidemiology, the proposals of this study were (1) to obtain the probability of occurrence of Oropouche, Mayaro, Saint Louis and Rocio fevers in Brazil based on environmental conditions corresponding to the periods of occurrence of the outbreaks; (2) to describe the macroclimatic scenario in Brazil in the last 50 years, evaluating if there was any detectable tendency to increase temperatures and (3) to model future expansion of those arboviruses in Brazil based on future temperature projections.

Methodology/Principal findings

Our model assessed seven environmental factors (annual rainfall, annual temperature, elevation, seasonality of temperature, seasonality of precipitation, thermal amplitude, and daytime temperature variation) for their association with the occurrence of outbreaks in the last 50 years. Our results suggest that various environmental factors distinctly influence the distribution of each arbovirus, with temperature being the central determinant of disease distribution in all high-risk areas. These areas are subject to change, since the average temperature of some areas has increased significantly over the time.

Conclusions/Significance

This is the first spatio-temporal study of the Oropouche, Mayaro, Saint Louis, and Rocio arboviruses, and our results indicate that they may become increasingly important public health problems in Brazil. Thus, next studies and control programs should include these diseases and also take into consideration key environmental elements.

(311805/2014-0). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

The Oropouche, Mayaro, Saint Louis, and Rocio viruses are neglected emerging mosquito-borne viruses that are spreading and causing wide-scale epidemics in South America. However, under-reporting of these cases is possible, as the symptoms are shared with other endemic diseases. Moreover, little is known regarding environmental conditions that favor these tropical outbreaks of arboviral diseases. This study examined the association of environmental factors with the probability of occurrence of Oropouche, Mayaro, Saint Louis and Rocio fever outbreaks (present and future) and finds that temperature is a central variable that determines the distribution of high-risk areas. This fact is very worrying, because the average temperature of some areas has increased significantly over the time. Results from this study strongly suggest that these four diseases have the potential to become important public health problems or become increasingly relevant in Brazil and other tropical areas in the coming years and should be monitored as part of effective control programs.

Introduction

Arboviruses have become important and constant threats in tropical regions, due to rapid climate change, deforestation, population migration, disorderly occupation of the urban areas, and precarious sanitary conditions that favor viral amplification and transmission [1]. Climate fluctuations produce conditions that accelerate arbovirus epidemics, directly affecting global public health [2]. Abnormally high temperatures for example, affect populations of insect vectors, and arboviral diseases, by influencing: the survival and replication of the virus, susceptibility of the vector to viruses, distribution of vectors, extrinsic incubation period of a virus in the insect, and seasonality of virus transmission patterns [3,4]. Besides that, arboviruses are highly spreadable because their vectors can be carried long distances, and even between countries or continents, which can lead to pandemics.

Brazil is the largest South American country and has a population of approximately 207 million in an area of 8,514,215 km² [5]. More than >30% of Brazil remains covered by rainforests and other natural ecosystems, despite the high rate of deforestation [5]. These natural environments can harbor many arboviruses that are maintained in different zoonotic cycles. For example, approximately 200 different arbovirus species have been isolated in Brazil, including 40 species that can cause human diseases [6,7]. Although it is acknowledged that dengue, zika, chikungunya, and recently, yellow fever, are today the most important emerging and re-emerging arboviral diseases in Brazil, in this study we focused on others that have been neglected and consequently, are less discussed in medical literature. These include Oropouche (ORO), Mayaro (MAY), Saint Louis (SLE), and Rocio (ROC). Besides the lack of specific tests to identify these diseases, the similarities among the symptoms are very high; fever, for example, is common to all of them. This makes the correct diagnosis very difficult and in most cases may have been underreported.

Oropouche (ORO)

The Oropouche virus (Orthobunyavirus genus) was first isolated in 1955 from a febrile human patient and *Coquillettidia venezuelensis* mosquitoes in Trinidad and Tobago [8]. Five years later, the virus was detected in a Brazilian territory in a sloth (*Bradypus tridactylus*) and in

Ochlerotatus serratus mosquitoes [9]. Since then, ORO has been a common cause of explosive urban epidemics in the Amazon region, affecting large cities such as Belem and Manaus.

This virus is transmitted among vertebrate hosts, such as marsupials, sloths, primates, and birds, through a generally wild transmission cycle by the *Ochlerotatus serratus* and *Culex quin-quefasciatus* mosquitoes. Notably, this arbovirus has adjusted to an urban transmission cycle with humans as the main reservoir and *Culicoides paraensis* (Ceratopogonidae) as the main vector. Thus, there is a worrisome risk of ORO emergence in the densely populated coast of Brazil, which cover the northeastern and southeastern regions, considering that vector *C. paraensis* is present in low-altitude areas of the entire Brazilian territory [7,10]. Moreover, *Cx. quin-quefasciatus* mosquitoes are spread throughout the Brazilian cities, suggesting the need to pay more attention to this mosquito species too.

ORO is one of the most important arboviral diseases in the Americas, especially in the Brazilian Amazon region. However, because ORO fever is not considered a reportable disease, it is difficult to estimate its incidence during outbreaks, although serological surveys are useful in this setting. Thus, research has indicated that approximately 500,000 people in the Amazon region may have been infected with the ORO virus since the early 1960s [6].

Most epidemics of ORO fever typically occur during the rainy season. However, some epidemics have also extended into the dry season, although with less intensity. The seasonal nature of the ORO is most likely linked to the higher density of the populations of the vector *C. paraensis* in months with higher levels of rainfall, combined with a higher concentration of exposed hosts. Unfortunately, the diagnosis of ORO can be confused with other acute febrile diseases that are endemic in the Amazon region, such as malaria and dengue [11].

Mayaro (MAY)

The MAY virus, belonging to the Alphavirus genus, has been responsible for outbreaks of acute febrile illness and arthralgia syndrome in northern and midwestern Brazil, as well as Peru, Bolivia, and Venezuela [7,12]. This virus was first detected and isolated in 1954, from rural workers in Trinidad [13].

Human cases of MAY are sporadic and mainly involve people who live in rainforests, as the main vector is the *Haemagogus* mosquitoes that are common in those forests. Vertebrate hosts are mainly mammals, although there is some evidence of bird infections in southern Brazil. *Aedes* mosquitoes can also transmit the virus in rural, suburban, and urban areas [14]. The course of 3–5 day of illness is characterized by fever, headache, myalgia, rash, and pain, mainly in the large joints, and less often, arthritis [15,16]. The spread of this virus can extend to cities through an infected human or through birds that can travel long distances in a short time, and adapt to a new cycle that involves humans as reservoirs.

This febrile illness occurs throughout the year, more frequently in the rainy season, as with dengue and ORO, and affects people of both sexes, of all ages. The estimated transmission of the virus in Manaus, state of Amazonas, is about 2 million people. This is a public health problem because there is no vaccine, and vector control is not feasible [12].

Saint Louis Encephalitis (SLE)

The SLE virus belongs to the Japanese encephalitis virus complex, which is within the *Flavivirus* genus, and Flaviviridae family [17]. The virus was first isolated in 1933from suspensions of human intracerebral brain samples that had been inoculated postmortem with tissues from rhesus monkeys and white mice (Saint Louis, Missouri, USA) [18]. Currently, the SLE virus is broadly distributed throughout all Americas (from Canada to Argentina), and has neurotropic characteristics [12]. It causes an acute disease in humans, with manifestations that range from febrile syndrome to fatal meningoencephalitis [19]. Reports of fatal cases vary from 5% to 20%; however, the numbers are even higher among the elderly population [20].

Transmission of the SLE virus occurs through *Culex* mosquitoes, and migratory birds spread the virus and other forms of encephalitis along their migratory routes [12]. Despite rare cases of the isolation of SLE virus in humans in Brazil, the antibodies of this virus were found in approximately 5% of the populations of the Northern and Southeastern Regions [12]. Recently, there was an outbreak of SLE in the country, which occurred simultaneously with that of dengue in São José do Rio Preto (Sao Paulo) [21]. During this outbreak, some patients with SLE exhibited hemorrhagic manifestations, such as a positive tourniquet test, petechiae, and bleeding [21].

Rocio (ROC)

The ROC virus was first isolated in 1975 from a fatal case of encephalitis in a restricted area of the Atlantic Forest (Ribeira River Valley Sao Paulo) [7]. The case was detected during the 1973–1980 outbreak which caused an estimated 1,000 cases of encephalitis in more than 20 municipalities. The mortality rate was 10%, and among the survivors, about 200 suffered balance or mobility sequelae [7]. It is unclear how the ROC virus spread to this region and why it subsequently disappeared 7 years later, although antibodies have been detected in rural residents of southeastern and northeastern Brazil. [22,23]. Based on the viral isolation and serological data, it is believed that the ROC virus is maintained in a transmission cycle that involves wild birds, including some migratory species, as the reservoirs, and *Aedes* and *Psorophora* mosquitoes, as the vectors.

Despite the availability of a comprehensive record in the literature for these relevant diseases, to the best of our knowledge, no predictive models have been developed in this context. In this study, we analyze and illustrate how these four mosquito-borne diseases can have serious public health implications, or increase their relevance in the future. Thus, the study's goals were: (a) to obtain the probability of occurrence of ORO, MAY, ROC and SLE in Brazil, based on environmental conditions corresponding to the periods of occurrence of the outbreaks; (b) to describe the macroclimatic scenario in Brazil in the last 50 years, evaluating any detectable tendency to increase temperatures and (c) to predict future expansion of ORO, MAY, SLE and ROC in Brazil, based on future temperature projections for 2046–2065 and 2071–2100, using two different scenarios of greenhouse gas emissions.

Methods

Study area and data source

The approximate locations of human ORO, MAY, SLE, and ROC cases were determined using sites that were identified in the literature between 1961 and 2012 (Table 1). Data were exhaustively collected using searches of the PubMed and Google Scholar databases (search term: "Oropouche" OR "Mayaro" OR "Saint Louis" OR "Rocio" AND "Brasil" OR "Brazil") and the library of University of Sao Paulo, Brazil. We have included all records of diseases in Brazilian municipalities reported in epidemiological bulletins since the very first record up until 2012. The criterion for inclusion of a municipality in the analysis was presence ≥ 1 of ORO, MAY, SLE or ROC case. This is because the World Health Organization has stated "*a single case of a communicable disease long absent from a population, or caused by an agent (e.g. bacterium or virus) not previously recognized in that community or area, or the emergence of a previously unknown disease, may also constitute an outbreak" [24].*

To determine the ecological and climatic conditions associated with ORO, MAY, SLE, and ROC mosquito-borne disease outbreaks, we examined the relationships between the locations

 Table 1. Brazilian municipalities that have presented arbovirus outbreaks in the 1961–2012 interim. (Data from Google Scholar and Pubmed databases. Searching topics = "Oropouche" OR "Mayaro" OR "Saint Louis" OR "Rocio" AND "Brasil" OR "Brazil"). The acronyms next to each municipality indicate the State.

NEGLECTED TROPICAL DISEASES

PLOS

:Ø

Virus	Municipality	Epidemic Year	References
Oropouche	Belém (PA)	1961, 1968, 1979, 1980	[25-29]
	Bragança (PA)	1967, 1979–1980	
	Baião (PA)	1972	
	Santarém region (PA)	1974–1975	
	Itupiranga (PA)	1975	
	Tomé Açu (PA)	1978	
	Portel (PA)	1979	
	Bragantina region (PA)	1979–1980, 2006	
	Mazagão (PA)	1980	
	Barcelos (AM)	1980	
	Manaus (AM)	1980–1981	
	Tocantinópolis (TO)	1988	
	Porto Franco (MA)	1988	
	Ouro Preto d'Oeste (RO)	1991	
	Ariquemes (RO)	1991	
	Serra Pelada (PA)	1994	
	Brasil Novo (PA)	1996	
	Novo Airão (AM)	1996	
	Oriximiná (PA)	1996	
	Vitória do Xingu (PA)	1996	
	Xapuri (AC)	1996	
	Parauapebas (PA)	2003	
	Porto de Moz (PA)	2004	
Mayaro	Belterra (PA)	1977–1978	[30-36]
	Conceição do Araguaia (PA)	1981	
	Itaruma (GO)	1987	
	Benevides (PA)	1991	
	Peixe (TO)	1991	
	Acrelândia (AC)	2004	
	Manaus (AM)	2007–2008	
	Santa Bárbara (PA)	2008	
	Sinop (MT)	2011–2012	
	Cuiabá (MT)	2012	
	Sorriso (MT)	2012	
	Várzea Grande (MT)	2012	
	Nossa Senhora do Livramento (MT)	2012	
Saint Louis	São Pedro (SP)	2004	[37-39]
	Ribeirão Preto (SP)	2006	_
	São José do Rio Preto (SP)	2007	
Rocio	Cubatão (SP)	1975	[40-43]
	Guarujá (SP)	1975	_
	Itanhaém (SP)	1975	_
	São Vicente (SP)	1975	_
	Mongaguá (SP)	1975	

(Continued)

Virus	Municipality	Epidemic Year	References
	Praia Grande (SP)	1975	
	Santos (SP)	1975	
	Cananéia (SP)	1975–1976	
	Iguape (SP)	1975–1976	
	Itariri (SP)	1975–1976	
	Jacupiranga (SP)	1975–1976	
	Juquiá (SP)	1975–1976	
	Miracatu (SP)	1975–1976	
	Pariquera-Açu (SP)	1975–1976	
	Pedro de Toledo (SP)	1975–1976	
	Peruíbe (SP)	1975–1976	
	Registro (SP)	1975–1976	
	Sete Barras (SP)	1975–1976	-
	Barra do Turvo (SP)	1976	
	Eldorado Paulista (SP)	1976]

Table 1. (Continued)

https://doi.org/10.1371/journal.pntd.0005959.t001

of the outbreaks and seven variables: annual rainfall (RAIN, mm), annual temperature (TEMP, °C), elevation (ELEV, m), seasonality of temperature (SEA-TEMP), seasonality of precipitation (SEA-RAIN), thermal amplitude (THER-AMP), and daytime temperature variation (DTV). The SEA-TEMP value was calculated as the standard deviation of the average monthly temperatures. The THERM-AMP value was calculated by subtracting the minimum temperature during the coldest month from the maximum temperature during the hottest month. The SEA-RAIN value was calculated as the coefficient of variation for average monthly precipitation. The mean DTV value was calculated by subtracting the mean minimum temperature from the mean maximum temperature. All weather data were obtained in ASCII-raster format files and using the "LAT/LONG" geodetic coordinate system (Datum WGS-84). These data were obtained from the WorldClim-Global Climate Data database, which contains representative observational data for 1950-2000 that were interpolated to a resolution of 30 arc-seconds (approximately 1 km). As the environmental variables were expressed in various units, the principal components analysis (PCA) was performed after standardizing the variables using a Pearson correlation matrix. The temperature layers for 1970-2010 were obtained from the National Institute of Meteorology [44], and data regarding other variables were obtained from the AMBDATA [45] and WorldClim [46] databases.

Data analysis

Our analysis included all probable (clinically diagnosed) and confirmed (serological) cases of persons with the onset of the disease from 1961 through to 2012. For each disease, the environmental variables analyzed were those that corresponded with the years of outbreaks: ORO (between 1961 and 2006), MAY (between 1977 and 2012), SLE (between 2004 and 2007) and ROC (1975 and 1976). The database was developed based on the presence and absence of arboviroses. We considered value **1** for years with at least one case (or more) of ORO, MAY, SLE or ROC, and value **zero** for other years (no occurrence), during the period studied (1961 to 2012). Table 1 shows the municipalities that had cases of these arboviruses and the years in which they occurred.

The PCA was performed using R software to preselect the environmental variables that had the greatest influence on the distributions of each disease [47]. The PCA approach was used for two reasons. First, PCA facilitates the identification and elimination of covariant variables, which is a key procedure for avoiding analytical artefacts. Second, PCA has been widely used in equivalent studies and then facilitates comparisons, reproducibility, and future meta-analysis. After the PCA, we selected the four most representative eigenvectors of the variables for each disease, which were used for Maxent analysis (version 3.3.3 k: a machine learning algorithm for modeling species distributions based on existing data and environmental variables) [48,49]. The data selection was performed according to the criterion of maximum entropy, with the original variables that reached maximum and minimum values within the ordered ranking of each principal component, because they describe the full range of data variation. The Maxent model may be expressed as:

$$p\Big[f_j\Big] = \frac{1}{N} \sum_{i=1}^{N} f_i(x_i)$$

Where: $x^* =$ the geographical region of interest; $x = \{x_1, x_2, ..., x_N\}$ with $x \in x^*$; $x \rightarrow$ observed points at x^* ; $f_j = f_1 \cdot, f_2 \cdot ..., f_m$ (environmental variables); N = the number of observed cases; and p = the probability of disease occurrence. The model was run 25 times, while withholding a difference of 10% of the localities for each run to estimate the parameters and its precision. The potential distribution maps were created by interpolating the occurrence points and the similarity measures of the environmental variables in each pixel (i.e., a known observation probability value can be assigned to each pixel by calculating a probability whose exponent is a quadratic function). To describe the temperature change patterns in Brazil during the sampled 50 years, we used the kriging method [50, 51] and data from approximately 250 monitoring stations throughout Brazil. This approach generated a map by estimating the value at each node of a regular grid, which was superimposed over the area of interest, and then a contouring program was applied to draw iso-level curves. We used a 250 × 250 grid of Brazil map, which provides 62,500 sections, because it was the maximum map resolution with a minimum required amount of computational time. R software [47] was also used to evaluate the temporal trend in temperature during the last five decades.

Future climate data were integrated using two global climate models (GCMs): the Had-GEM2-ES [52] and MIROC-5 [53], which were selected for their different strengths. The Had-GEM2-ES model is a stable model that represents a realistic state of the climate, vegetation, and oceanic biology, without the need for artificial corrections. On the other hand, the MIROC-5 model also includes components of the Earth's system and climate change, in relation to anthropogenic radiation. The advantage of using this model is that it increases the accuracy of short-term climate prediction, as it can be affected by both anthropogenic and intrinsic fluctuations of the climate system. The spatial resolution of the GCMs was the same as that of the environmental variables (30 arc-seconds, approximately 1 km). The comparison method was the same as for the Maxent model, although the probability calculation for the GCMs incorporated a comparison of the present and future environmental conditions. To obtain future climate scenarios using GCMs, it is also necessary to choose a condition for evolution of the greenhouse gas emissions (GGE), during the period when the future climate is projected. In our prediction we used two different scenarios: low emission (RCP 2.6) and very high emission (RCP 8.5), detailed in the Special Report on Emissions Scenarios by the Intergovernmental Panel on Climate Change [54]. In the first case, the global temperature tends to increase by 1.0°C and can reach a temperature anomaly ranging from 0.4 to 1.6°C and 0.3 to 1.7°C between 2046–2065 and 2081–2100, respectively [55]. In the second scenario, with high GGE,

the global temperature tends to increase 2.0 to 3.7°C and can reach to a thermal anomaly ranging from 1.4 to 2.6°C and 2.6 to 4.8°C between 2046–2065 and 2081–2100, respectively [56,57].

The models of future expansion of ORO, MAY, SLE and ROC in Brazil were then projected into the timeline, and the two future climatic conditions (2046–2065 and 2071–2100), to identify areas suitable for those diseases. A map of raw temperature projections from the GCMs used to drive the disease models can be seen at <u>S1 Fig</u>. The default Maxent auto feature setting was used (linear, quadratic, product, threshold, and hinge). The maps were edited using QGis software 2.10.1.

Results

Through PCA of climatic factors, it was possible to identify three main groups: ROC, SLE and ORO + MAY (Fig 1). It is important to note that both diseases ORO and MAY occurred more in the North and Midwest of the country. The first two components (F1 and F2) were able to explain 82.96% of the variation.

Analyzing each disease separately (Fig 2), according to PCA it was possible to perceive that the most influential factors were distinct for each arbovirus. With respect to ORO cases, the most important variables were TEMP and SEA-TEMP; for MAY: THERM-AMP and SEA-TEMP, which was similar to ORO; for SLE: RAIN and DTV, and finally for ROC, the most important variables were THERM-AMP and ELEV. Details are described in Table 2.

As some variables co-varied (Fig 2), we selected only the non-covariant variables as input for analysis in Maxent software. The cut-off was four variables and was based on ROC, which presented the lowest number (four) of non-covariant variables. After selecting the four most important variables for each disease, we constructed a predictive model in Maxent (Fig 3), in order to determine what areas were most likely to present outbreaks. The contribution of each variable for each model is described in Table 3. The final model for ORO, MAY, SLE and ROC had an area under the curve of 0.79, 0.76, 0.85 and 0.99, respectively, significantly better than the random prediction (p = 0.001), indicating good performance of the model. The Maxent outputs and receiver operating characteristic curves [58] for all arboviruses are shown in S2 Fig. We observed that there is a concentration of ORO and MAY in the Northern region of Brazil, while SLE and ROC are mainly present in the South region and coastal region.





https://doi.org/10.1371/journal.pntd.0005959.g001



Fig 2. The most influential environmental variables. PCA of each disease showing which environmental variables are the most influential.

https://doi.org/10.1371/journal.pntd.0005959.g002

Most of the important variables for the distribution of all four diseases are temperaturerelated (TEMP, SEA-TEMP); so we analyzed the temperature situation in Brazil in the last 50 years. After analyzing the historical temperature series, and using the kriging method, we realized that there has been an increase in temperature over the decades (Fig 4), especially in the North of the country.

In the case of continuity of this scenario of temperature increase, we generated probability maps with two different climate future projections (Fig 5). The results reveal a progressively expanding areas with an increased likelihood of ORO, MAY, SLE and ROC cases, especially at the edges of the transmission areas. In scenario of high GGE it was possible to observe the increase of high risk areas for ORO and MAY, while for SLE and ROC there were no drastic changes. This fact is in agreement with our observations of temperature increase (Fig 4),

	Eigenvectors							
Environmental Variables	ORO		MAY		SLE		ROC	
	F1	F2	F1	F2	F1	F2	F1	F2
ТЕМР	0.461	0.006	-0.375	0.011	0.949	-0.316	0.129	-0.672
SEA-RAIN	-0.259	-0.156	0.369	-0.415	0.955	-0.298	0.382	-0.127
RAIN	0.407	-0.135	-0.352	-0.510	0.046	0.999	-0.459	0.068
ELEV	-0.394	0.040	0.387	-0.142	-0.741	0.672	0.039	0.705
DTV	-0.428	-0.311	0.397	-0.366	0.990	0.144	0.460	0.109
THER-AMP	-0.448	-0.025	0.421	-0.089	-0.735	-0.678	0.467	0.087
SEA-TEMP	-0.126	0.926	0.339	0.637	-0.898	-0.440	0.441	0.109

Table 2. Importance of environmental variables according to disease.

Values of PCA showing the relative influence of which environmental variables for each disease. The most influential variables (extreme values in bold) were used in the later analyzes with Maxent software.

https://doi.org/10.1371/journal.pntd.0005959.t002

which the greatest changes occurred precisely in the North region of the country, affecting mainly the distribution of ORO and MAY. We also performed these same analyzes with the GCM MIROC-5 and the results were essentially the same (S3 Fig).



Fig 3. Occurrence of arboviruses. Map of the Brazilian territory showing probability of areas with ORO (AUC: 0.79), MAY (AUC: 0.76), SLE (AUC: 0.85) and ROC (AUC: 0.99) during outbreaks in the last five decades. The maps were built using QGis software 2.10.1.

https://doi.org/10.1371/journal.pntd.0005959.g003

Environmental Variables	Contribution (%)					
	ORO	MAY	SLE	ROC		
TEMP	1.2	92.3	-	50.7		
SEA-RAIN	-	-	-	-		
RAIN	-	2.2	11.5	13.6		
ELEV	-	-	-	25.8		
DTV	3.6	-	1.4	-		
THER-AMP	2.4	5.5	24.5	9.9		
SEA-TEMP	92.8	0	62.7	-		
AUC index	0.79	0.76	0.85	0.99		

Table 3. Percent contribution of each variable in the final Maxent model.

Percent contribution of each variable chosen for the final model obtained with Maxent. The dashes indicate those variables excluded (owing to co-variation or low influence).

https://doi.org/10.1371/journal.pntd.0005959.t003

Discussion

Modeling outbreaks in Brazil

Our analysis showed that the occurrence of outbreaks of ORO, MAY, SLE and ROC is affected differently by environmental variables, although temperature seems to be strongly associated with all of them. Theses information is quite alarming, since the temporal analyses have shown that the average temperature of some Brazilian areas has been increasing over the decades (see Fig 4). This variable presents an important constraint on the extent of expansion of these diseases throughout the country, mainly because these changes are intimately linked to vector





https://doi.org/10.1371/journal.pntd.0005959.g004



Fig 5. Predicted ORO, MAY, SLE and ROC range expansion in Brazil based on GCM HadGEM2-ES. The maps show the distribution under two climate change scenarios: RCP 2.6 (lower increase in greenhouse gas emissions) and RCP 8.5 (higher increase in greenhouse gas emissions). The maps were built using QGis software 2.10.1. *"Present" is the scenario in which disease outbreaks have been described, based on 1950–2010 climate data.

https://doi.org/10.1371/journal.pntd.0005959.g005

lifecycle development and associated with the virus itself [7,59]. In addition, the IPCC report [60] points out that heat waves are more prone to occur in the next years with more frequency and duration. Furthermore, events of extreme precipitation are also most likely to be more frequent and intense on the continental surfaces, in humid tropical regions such as the North of Brazil. Consequently, climate changes may directly interfere with the distribution of the diseases evaluated in this study, particularly ORO and MAY.

The increase in temperature may also change the distribution of virus vectors, because they may migrate to other areas where conditions are favorable for proliferation. In some cases, the elevation has an inverse effect on the temperature, because generally, higher altitudes correspond with lower temperatures; nevertheless, some vectors are adaptable. In Mexico, the vector *Aedes aegypti* for example, has been found at an elevation of 2,000 meters whereas previously, it was only found in places up to 1,000 meters high [61]. For ORO and MAY

outbreaks specifically, one of the factors that seems to be more related is environmental change [59]. Following deforestation in the Amazon, and subsequently, the cocoa plantation and others cultivated in the region, the vectors found an ideal spot for reproduction in the cocoa shells, and, expanded their population, spreading the virus to humans along the Amazonian roads [62]. Therefore, ORO and MAY outbreaks appear to be the result of a strong relationship between the virus and its environment, with human activity (colonization, cocoa cultivation, and subsequent environmental changes) resulting in the proliferation of the Culicidae and Ceratopogonidae mosquito families, and subsequently increasing their human contact [59,62,63].

By analyzing the results of PCA, it was observed that diseases were grouped into three main clusters: SLE, ROC and ORO+MAY (see Fig 2). The relationship between ORO and MAY occurred because both diseases are influenced by similar climatic factors, such as seasonality of temperature. Both have high-risk areas in the Northern and Central regions of the country, where the Amazonian biome predominates [5]. Moreover, both diseases are related to wild vectors, such as *Haemagogus* mosquitoes for cases of MAY and *C. paraensis* (Ceratopogonidae) for ORO [7]. Furthermore, although restricted to the Amazon region, these diseases have the potential to spread throughout the country and around the world, since they affect birds that can move long distances by migration or by illegal wildlife trafficking, which is very common in this region.

In relation to ROC, the factors that were most related to high-risk areas were thermal amplitude and elevation. This is in accordance with the distribution of the Atlantic Forest biome, which covers a mountainous region that shelters the wild vectors, *Aedes scapularis* and *Psorophora sp.* mosquitoes. To the best of our knowledge, no reports have explained the emergence and sudden disappearance of ROC cases in the 1970s, but it is believed that this virus is remained in a cycle where birds, including some migratory species, are the reservoirs [12]. Furthermore, in 2004, antibodies were detected among birds in southern Brazil, and the ROC virus may be circulating in different Brazilian regions, which could represent a permanent threat of disease outbreaks [7,12].

Similarly, the SLE virus can also be found in migratory birds, and it is believed to be responsible for the spread of the virus throughout the Americas, as well as other encephalitis. However, there are biological and genetic differences between isolates from North and South America [64]. Despite the rare isolates of SLE in humans in Brazil, antibodies to this virus were found in approximately 5% of the populations of the Northern and Southeastern regions [21]. The present study revealed that the most important factors for SLE were annual RAIN and DTV, which can also affect the life cycle of the *Culex* vector. For example, greater rainfall results in increasing vegetation coverage, which is the primary food source of many vertebrates that are potential hosts for the mosquitoes. However, decreased precipitation can reduce vegetation and drive both the vertebrate hosts and the mosquitoes towards human settlements, which can increase vector-human contact [2,3,65].

By analyzing the distribution maps of the diseases generated by Maxent it was observed that the probable high-risk areas were much larger than those in which the cases were detected. For example, ROC cases were reported only in the Sao Paulo state, but it is possible that there are other areas suitable for other outbreaks, such as a small region observed between the Rondonia and the Mato Grosso states (see Fig 3). This discrepancy between the reported cases and the existing cases is due mainly to the similarity of the clinical symptoms among those arboviruses, leading to an underestimation of the occurrence of ORO, MAY, SLE and ROC in Brazil. This scenario is aggravated by the lack of accurate diagnostic methods that identify which virus is acting. In cases of acute febrile illness outbreaks of MAY, SLE and ORO, for example, the laboratory procedures for diagnosing suspected cases is indispensable because these pathogens cannot be differentiated from other viral diseases, such as dengue or chikungunya, and may remain unknown [7].

As Brazil has been facing a major dengue, zika and chikungunya epidemic in recent years [66], it is quite possible that other arboviruses with similar symptoms have been underreported or confused with dengue itself. Despite the knowledge of the significant occurrence of many arboviruses in the Amazon Region, like ORO and MAY, many cases remain undiagnosed, probably because of their clinical manifestations, being usually mild and self-limited; patients generally recover completely after a few days. However, more severe cases may remain undiagnosed, especially because of the long distances to health care facilities, transport difficulties of the sample and lack of laboratories capable of conducting diagnostic tests. With regard to ORO infections, the diagnosis can easily be confused with malaria, which is endemic in that region [11,12].

According to the predictive model HadGEM2-ES, high-risk areas for all diseases may change in the next decades. For MAY, for example, practically the whole country will have adequate climatic conditions for virus transmission (see Fig 5). In some cases, such as ORO, total areas of infection will decline, but those that remain, such as the Northwestern and Northeastern regions of the country will be at increased risk. For ROC, there will be an emergent highrisk area in the South of the country, while for SLE the southern areas will become less susceptible to outbreaks. This scenario can be more or less dramatic, and depends on the levels of greenhouse gas emissions. However, this difference in emission rates will most strongly affect ORO and MAY distribution, precisely because they occupy the Northern areas of Brazil and are more susceptible to temperature changes (see Fig 4). Climatic change also affect human activity and the migration of the people, as well as the redistribution of the vectors, and a more favorable environment for the propagation of arboviruses [59]. Therefore, future risk estimations should consider those factors. Nevertheless, despite all efforts and cumulated knowledge in the literature, it is still remains difficult to identify the main cause of an outbreak [67] and to determine the most efficient technique(s) for protecting humans from these viruses. However, our findings indicate that temperature-related variables appear to play a central role in the epidemiology of culicid-vectored arboviruses.

Limitations

This study has limitations, since the projections presented here are processed on the assumption that the other variables remained stable. The human population size, for example, was considered as constant in our models. Moreover, we did not consider host migration (of both birds and humans), and the deforestation rate, which are important factors in determining the outbreaks of ORO and SLE. Yet, other factors that were not part of our model could change over the given time period: For example, the quality of vector surveillance, clinical case detection, and the development of some vaccine. Besides that, we did not take into account the distribution areas of the vectors, which are essential for the transmission, but very difficult to estimate. Another point is we used linear extrapolation to predict the distributions of each disease, and our model did not account for non-linear phenomena. For example, the models would fail if temperature increased to the point that it changed the vector transmission rates.

Another limitation is the supposed underreporting of cases and the confusion with other diseases. These neglected arboviruses may be circulating in asymptomatic patients or misdiagnosed and it is not possible to know their exact distribution; this affects directly the predictive power of our model. The model itself also has an intrinsic limitation, because it assumes that the increase in temperature can only be owing to GGE. We know that this is not the only possibility in nature, so this is a simplified premise that limitates interpretations. Although there are a number of constraints to our model, it is an important first step in trying to predict the emergence of these neglected arboviruses. It also serves to warn the vigilant surveillance health that arboviruses may remain longly underestimated before outbreak arises.

Conclusions

In summary, environmental factors can directly affect the distribution of ORO, MAY, SLE and ROC. Among them, temperature is a central variable that determines the distribution of highrisk areas. As the average temperature of some Brazilian areas has augmented significantly over the last 50 years, a better understanding of the biology of neglected arboviruses, their interactions and consequences in the ecosystem and climatic factors is needed. The four diseases addressed in the present work are clearly a latent menace to the public health and thus should be promptly included in the health programs agenda. Accurate detection and diagnosing are fundamental steps for developing efficient control measures. These four apparently similar arboviruses are differently affected by environmental factors, and these differences are probably linked to the vector's lifecycle or associated with the virus itself. We can also conclude that our mathematical and statistical approach allowed us to further describe peculiar environmental elements in the epidemiology of these neglected diseases. Even though, the approach is limited and we suggest that next studies should be multidisciplinary and comprehensive so that they include vector distribution, components of natural cycle and actual disease incidence.

Overall, this work provides useful indications about the dynamics of those arboviruses across the country. Our results suggest that high-risk areas may change in the coming years, being more pronounced with high GGE rates in the northern region of the country. This is the first spatio-temporal study of these arboviral diseases. However, there are gaps in available knowledge to scientifically predict future occurrences of large epidemics. Due to the epidemio-logical and entomological situation of several continents, there is evidence of aggravation of this current scenario, because there is great difficulty in eliminating or controlling the risk factors of the diseases.

Supporting information

S1 Fig. Raw temperatures. Maps of raw temperature projections from the GCMs models. (TIF)

S2 Fig. Maxent outputs. Detailed output data and receiver operating characteristic (ROC) curves for each disease.

(PDF)

S3 Fig. Predicted ORO, MAY, SLE and ROC range expansion in Brazil based on GCM MIROC-5. The maps show the distribution under two climate change scenarios: RCP 2.6 (lower increase in greenhouse gas emissions) and RCP 8.5 (higher increase in greenhouse gas emissions). The maps were built using QGis software 2.10.1. *"Present" is the scenario in which disease outbreaks have been described, based on 1950–2010 climate data. (TIF)

Author Contributions

Conceptualization: Camila Lorenz, Flávia Virginio, Francisco Chiaravalloti-Neto, Lincoln Suesdek.

Data curation: Camila Lorenz, Thiago S. Azevedo, Flávia Virginio.

Formal analysis: Thiago S. Azevedo, Francisco Chiaravalloti-Neto, Lincoln Suesdek.

Investigation: Camila Lorenz, Thiago S. Azevedo, Flávia Virginio, Francisco Chiaravalloti-Neto, Lincoln Suesdek.

Methodology: Thiago S. Azevedo, Francisco Chiaravalloti-Neto.

Project administration: Camila Lorenz, Francisco Chiaravalloti-Neto, Lincoln Suesdek.

Software: Thiago S. Azevedo.

Supervision: Francisco Chiaravalloti-Neto, Lincoln Suesdek.

Validation: Thiago S. Azevedo.

Visualization: Breno S. Aguiar.

Writing - original draft: Camila Lorenz, Flávia Virginio.

Writing – review & editing: Camila Lorenz, Thiago S. Azevedo, Flávia Virginio, Breno S. Aguiar, Francisco Chiaravalloti-Neto, Lincoln Suesdek.

References

- 1. Rust RS. Human arboviral encephalitis. Seminar Ped Neurol. 2012. 130–151.
- Anyamba A, Linthicum KJ, Small JL, Collins KM, Tucker CJ, Pak EW, Russell KL. Climate teleconnections and recent patterns of human and animal disease outbreaks. PLoS Negl Trop Dis. 2012; 6(1), e1465. https://doi.org/10.1371/journal.pntd.0001465 PMID: 22292093
- 3. Gubler DJ. Human arbovirus infections worldwide. Ann NY Acad Sci. 2001; 951(1): 13–24.
- Epstein PR. Climate change and human health. N Engl J Med. 2005; 353(14):1433–36. https://doi.org/ 10.1056/NEJMp058079 PMID: 16207843
- 5. Instituto Brasileiro de Geografia e Estatística. 2015. Available from: http://www.ibge.gov.br/
- 6. Rosa APAT, Vasconcelos PFC, Rosa JFST. An overview of arbovirology in Brazil and neighbouring countries. Instituto Evandro Chagas Press, 1998: 19–31.
- 7. Figueiredo LTM. Emergent arboviruses in Brazil. Rev Soc Bras Med Trop. 2007; 40(2): 224–29. PMID: 17568894
- Anderson CR, Spence L, Downs WG, Aitken TH. Oropouche virus: a new human disease agent from Trinidad, West Indies. Am J Trop Med Hyg. 1961; 10(4), 574–578.
- Pinheiro FM, Pinheiro G, Bensabath OR, Shope R. Epidemia de virus oropouche em Belém. Rev Ser Esp Saude Pub. 1962; 12:15–23.
- Felippe-Bauer ML, Sternheim US. *Culicoides paraensis* (Diptera: Ceratopogonidae) infestations in cities of the Itapocu river valley, southern Brazil. Entomol News. 2008; 119(2), 185–192.
- Mourão MP, Bastos MS, Gimaque JB, Mota BR, Souza GS, Grimmer GH, et al. Oropouche Fever Outbreak, Manaus, Brazil, 2007–2008. Emerg Infect Dis. 2009; 15(12): 2063–64. <u>https://doi.org/10.3201/eid1512.090917</u> PMID: 19961705
- Lopes N, Nozawa C, Linhares REC. Características gerais e epidemiologia dos arbovírus emergentes no Brasil. Rev Pan Amaz Saude. 2014; 5(3): 55–64.
- Anderson CR, Downs WG, Wattley GH, Ahin NW, Reese EAA. Mayaro virus: a new human disease agent. II. Isolation from blood of patients in Trinidad, B.W.I. Am J Trop Med Hyg. 1957; 6: 1012–1016. PMID: 13487973
- 14. Muñoz M, Navarro JC. Mayaro: a re-emerging arbovirus in Venezuela and Latin America. Biomed. 2012; 32(2): 286–302.
- Gould EA. Understanding the alphaviruses: Recent research on important emerging pathogens and progress towards their control. Antiviral Research. 2010; 87(2): 111–124. https://doi.org/10.1016/j. antiviral.2009.07.007 PMID: 19616028
- Pinheiro FP, Leduc JW 1998. Mayaro fever. In Monath TP (ed.), The arboviruses: epidemiology and ecology, Vol. III, CRC Press, Boca Ratón, p. 137–150.

- Calisher CH, Karabatsos N, Dalrymple JM, Shope RE, Porterfield J, Westaway EG, Brandt WE. Antigenic relationships between flaviviruses as determined by cross neutralization tests with polyclonal antisera. J. Gen. Virol. 1989; 70: 37–43. https://doi.org/10.1099/0022-1317-70-1-37 PMID: 2543738
- Reisen WK. Epidemiology of St. Louis encephalitis virus. Advances in virus research. 2003; 61: 139– 184. PMID: 14714432
- Vasconcelos PF, Travassos Da Rosa JFS, Travassos Da Rosa APA, Degallier N, Pinheiro FP, Sá Filho GC. Epidemiologia das encefalites por arbovirus na Amazônia brasileira. Rev Inst Med Trop São Paulo. 1991; 33:465–76. PMID: <u>1844977</u>
- Hull R, Nattanmai S, Kramer LD, Bernard KA, Tavakoli NP. A duplex real-time reverse transcriptase polymerase chain reaction assay for the detection of St. Louis encephalitis and eastern equine encephalitis viruses. Diagn Micr Infec Dis. 2008; 62(3): 272–279.
- Mondini A, de Morais Bronzoni RV, Cardeal ILS, Dos Santos TMIL, Lázaro E, Nunes SHP, Neto FC. Simultaneous infection by DENV-3 and SLEV in Brazil. J Clin Virol. 2007; 40(1): 84–6. <u>https://doi.org/10.1016/j.jcv.2007.06.007</u> PMID: 17658293
- Figueiredo LTM, da Rosa APT, Fiorillo AM. Níveis de anticorpos para arbovírus em indivíduos da região de Ribeirão Preto, SP (Brasil). Rev Saude Publ. 1986; 20(3): 204–211.
- Straatmann A, Santos-Torres S, Vasconcelos PF, Travassos da Rosa A, Rodrigues SG, Tavares-Neto J. Evidências sorológicas da circulação do arbovírus Rocio (Flaviviridae) na Bahia. Rev Soc Bras Med Trop. 1997; 30(6): 511–515. PMID: 9463199
- 24. World Health Organization. Health Topics: Diseases Outbreaks. Available from: http://www.who.int/ topics/disease_outbreaks/en/
- Watts DM, Ramirez G, Cabezas C, Wooster MT, Carrillo C, Chuy M, Gentrau EJ, Hayes CG. Arthropod-borne viral diseases in Peru. Instituto Evandro Chagas Press, 1998; 193–218.
- da Silva Azevedo RDS, Nunes MRT, Chiang JO, Bensabath G, Pinto N, Martins L, da Costa Vasconcelos PF. Reemergence of Oropouche fever, northern Brazil. Emerg Infect Dis. 2007; 13(6): 912. https:// doi.org/10.3201/eid1306.061114 PMID: 17553235
- 27. Nunes M. Oropouche Virus Isolation, Southeast Brazil. Emerg Infect Dis. 2005; 10(11).
- Terzian ACB. Sporadic Oropouche Infection, Acre, Brazil. Emerg Infect Dis. 2009; 15(2): 348–350. https://doi.org/10.3201/eid1502.080401 PMID: 19193295
- 29. Vasconcelos HB. Molecular Epidemiology of Oropouche Virus, Brazil. Emerg Infect Dis. 2011; 17(5).
- Hoch AL, Peterson NE, LeDuc JW, Pinheiro FP. An outbreak of Mayaro virus disease in Belterra, Brazil. III. Entomological and ecological studies. Am J Trop Med Hyg. 1981; 30(3): 689–698. PMID: 6266265
- Halsey ES, Siles C, Guevara C, Vilcarromero S, Jhonston EJ, Ramal C, Ampuero JS. Mayaro virus infection, Amazon basin region, Peru, 2010–2013. Emerg Infect Dis. 2013; 19(11): 1839–1842. <u>https:// doi.org/10.3201/eid1911.130777</u> PMID: 24210165
- Zuchi N, Heinen LBDS, Santos MAMD, Pereira FC, Slhessarenko RD. Molecular detection of Mayaro virus during a dengue outbreak in the state of Mato Grosso, Central-West Brazil. Mem I Oswaldo Cruz. 2014; 109(6): 820–823.
- Vieira CJDSP, da Silva DJF, Barreto ES, Siqueira CEH, Colombo TE, Ozanic K, de Morais Bronzoni RV. Detection of Mayaro virus infections during a dengue outbreak in Mato Grosso, Brazil. Acta Trop. 2015; 147: 12–16. https://doi.org/10.1016/j.actatropica.2015.03.020 PMID: 25817238
- Patiño-Barbosa AM, Bedoya-Arias JE, Cardona-Ospina JA, Rodriguez-Morales A J. Bibliometric assessment of the scientific production of literature regarding Mayaro. J Infection Pub H. 2016; 9(4): 532–534.
- Receveur MC, Grandadam M, Pistone T, Malvy D. Infection with Mayaro virus in a French traveller returning from the Amazon region, Brazil, January, 2010. Euro Surveill. 2010; 15(18): 195–263.
- 36. Azevedo RS. Mayaro Fever Virus, Brazilian Amazon. Emerg Infect Dis. 2009; 15(11).
- Rocco IM, Santos CL, Bisordi I, Petrella SM, Pereira LE, Souza RP, Cerroni MP. St. Louis encephalitis vírus: first isolation from a human in São Paulo state, Brasil. Rev Inst Med Trop. 2005; 47(5): 281–285.
- Heinen LBDS, Zuchi N, Serra OP, Cardoso BF, Gondim BHF, Santos MAMD, Dezengrini-Slhessarenko R. Saint Louis encephalitis virus in Mato Grosso, central-western Brazil. Rev Inst Med Trop. 2015; 57 (3): 215–220.
- Terzian ACB, Mondini A, de Moraes Bronzoni RV, Drumond BP, Ferro BP, Cabrera SEM, Nogueira ML.Detection of Saint Louis encephalitis virus in dengue-suspected cases during a dengue 3 outbreak. Vector-Borne Zoo Dis. 2011; 11(3): 291–300.
- 40. Iversson LB. Epidemia de encefalite por arbovírus na região sul do estado de São Paulo, Brasil, em 1975 e 1976. Aspectos da distribuição cronológica e geográfica dos casos. Rev Saúde Pub. 1977; 11: 375–388.

- Iversson LB. Aspectos da epidemia de encefalite por arbovírus na região do vale do Ribeira, São Paulo, Brasil, no período 1975–1978. Rev Saúde Pub. 1980; 14: 9–35.
- Lopes ODS, Coimbra TL, Sachhetta LDA, Calishier CH. Emergence of a new arbovirus disease in Brazil: Isolation and characterization of the etiologic agent, Rocio Virus. Am J Epidemio. 1978; 107(5): 444–449.
- 43. Iversson LB, Coimbra TLM. Encefalite na região do Vale do Ribeira, São Paulo, Brasil, no período pósepidêmico de 1978 a 1983: Situação do diagnóstico etiológico e características epidemiológicas. Rev Saúde Pub. 1984; 18(4): 323–332.
- 44. Instituto Nacional de Meteorologia. BDMEP—Banco de Dados Meteorólogicos para Ensino e Pesquisa. 2017. Available from: http://www.inmet.gov.br/portal/index.php?r=bdmep/bdmep
- Amaral S, Costa CB, Arasato LS, Ximenes AC, Rennó CD. AMBDATA: Variáveis ambientais para Modelos de Distribuição de Espécies (SDMs). Available from: http://urlib.net/3ERPFQRTRW34M/3E7GH36.
- Hijimans RJ, Camaron S, Parra J, Jones PG, Jarvis A. Very high resolution interpolated climate surfaces for global land areas. Int J Climatol. 2005; 25: 1965–1978.
- R Development Core Team. R: A Language and Environment for Statistical Computing. ISBN 3-900051-07-0. R Foundation for Statistical Computing, Vienna, Austria. 2008.
- Phillips SJ, Dudík M. Modeling of species distributions with Maxent: new extensions and a comprehensive evaluation. Ecography. 2008; 31(2): 161–175.
- Elith J, Phillips SJ, Hastie T, Dudík M, Chee YE, Yates CJ. A statistical explanation of MaxEnt for ecologists. Div Dist. 2011; 17(1): 43–57.
- Krige DG. A statistical approach to some mine evaluations and allied problems on the Witwatersrand. Master's thesis, University of the Witwatersrand, Johannesburg, South Africa, 1951.
- 51. Carrat F, Valleron AJ. Epidemiologic mapping using the "kriging" method: application to an influenzalike epidemic in France. Am J Epidem. 1992; 135(11): 1293–1300.
- Collins WJ, Bellouin N, Doutriaux-Boucher M, Gedney N, Halloran P, Hinton T, Martin G. Development and evaluation of an Earth-System model-HadGEM2. Geosci Model Dev Discuss. 2011; 4(4): 1051.
- Watanabe S, Hajima T, Sudo K, Nagashima T, Takemura T, Okajima H, Ise T. MIROC-ESM: model description and basic results of CMIP5-20c3m experiments. Geosci Model Dev Discuss. 2011; 4(2): 1063–1128.
- Pachauri RK, Meyer L, Plattner GK, Stocker T. IPCC, 2014: Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. 2015.
- Van Vuuren DP, Den Elzen MG, Lucas PL, Eickhout B, Strengers BJ, Van Ruijven B, van Houdt R. Stabilizing greenhouse gas concentrations at low levels: an assessment of reduction strategies and costs. Climatic Change. 2007; 81(2): 119–159.
- 56. Chou SC, Lyra A, Mourão C, Dereczynski C, Pilotto I, Gomes J, Bustamante J, Tavares P, Silva A, Rodriguez D, Campos D, Chagas D, Sueiro G, Siqueira G, Marengo J. Assessment of Climate Change over South America under RCP 4.5 and 8.5 Downscaling Scenarios. Am J Clim Change. 2014; 3: 512– 525.
- Riahi K, Gruebler A, Nakicenovic N. Scenarios of long-term socio-economic and environmental development under climate stabilization. Technol Forecast Soc. 2007; 74(7): 887–935.
- **58.** Fawcett T. An introduction to ROC analysis. Pattern Recognition Letters. 2006; 27(8): 861–874.
- 59. Pignatti MG. Saúde e ambiente: as doenças emergentes no Brasil. Amb Soc. 2004; 7(1): 133–44.
- **60.** IPCC. Summary for Policymakers in Climate Change 2014: Mitigation of Climate Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (eds Edenhofer O. et al.) Cambridge Univ. Press, 2014.
- Harrera-Bastos E, Prevots DR, Zarate ML, Sila JL, Sepulveda-Amor J. First reported outbrek of classical dengue fever at 1.700 meters above sea level in Guerrero State, Mexico. Am J Trop Med Hyg. 1992; 46(6): 649–653. PMID: 1621889
- 62. Waldaman EA, Silva LJ, Monteiro CA. Trajetória das doenças infecciosas:da eliminação da poliomelite à reintrodução do cólera. São Paulo, Hucitec/NUPEN/USP Press, 1995: 195–244.
- 63. Gibbons A. Where are "new" diasease born? Science. 1993; 261(6).
- Vasconcelos HB, Azevedo RSS, Casseb SM, Nunes-Neto JP, Chiang JO, Cantuária PC, et al. Oropouche fever epidemic in northern Brazil: epidemiology and molecular characterization of isolates. J Clin Virol. 2009; 44:129–133. https://doi.org/10.1016/j.jcv.2008.11.006 PMID: 19117799
- Zhou G, Flowers M, Friedrich K, Horton J, Pennington JE, Wells MA. Metabolic fate of [14C]-labeled meal protein amino acids in *Aedes aegypti* mosquitoes. J Insect Physio. 2004; 50: 337–349.

- 66. Ministério da Saúde. Boletim Epidemiológico. 2015. Available from: http://portalsaude.saude.gov.br/
- 67. Massad E. The Risk of Chikungunya Fever in a Dengue-Endemic Area. J Travel Med. 2008; 15: 147– 155. https://doi.org/10.1111/j.1708-8305.2008.00186.x PMID: 18494691